Author's response to reviews

Title: Toona Sinensis Leaf Extract Has Anti-nociceptive Effect Comparable to Non-steroid anti-inflammatory Agents in Mouse Writhing Test

Authors:

Yu-feng Su (suyufeng2000@gmail.com)
YU-CHIAO Yang (m755016@kmu.edu.tw)
HSENG-KUANG Hsu (m585004@yahoo.com.tw)
Shiuh-Lin Hwang (nsdoctor@yam.com)
Kung Shing Lee (leekungshing@yahoo.com.tw)
Ann-Shung Lieu (e791125@gmail.com)
Te-fu Chan (tefu.chan@msa.hinet.net)
Chih-Lung Lin (chihlung1@yahoo.com)

Version: 2 Date: 12 August 2014

Author's response to reviews:

1. I've added the line and page numbers in the manuscript.
2. The name of staff who undertook the formal identification of the plant material, the voucher specimen number and the place the plant has been deposited are provided and added in the manuscript.